Author's response to reviews

Title: Swimming performances in long distance open-water events with and without wetsuit

Authors:

Sebastian Ulsamer (ulsamer1@yahoo.de)
Christoph A Rüst (christoph.ruest@bluewin.ch)
Thomas Rosemann (thomas.rosemann@usz.ch)
Romuald Lepers (romuald.lepers@u-bourgogne.fr)
Beat Knechtle (beat.knechtle@hispeed.ch)

Version: 4 Date: 11 May 2014

Author's response to reviews:

PD Dr. med. Beat Knechtle
Facharzt FMH für Allgemeinmedizin
Holengartenstr. 3e
9300 Wittenbach
Switzerland
beat.knechtle@hispeed.ch

Wittenbach, May 11, 2014

BMC Sports Science, Medicine and Rehabilitation
Prof. Dr. Fernando Marques
Editor-in-Chief

Dear Prof. Marques

MS: 1168184054112245
Swimming performances in long distance open-water events with and without wetsuit

We thank you for your e-mail from May 9, 2014.

As suggested, we went again through the manuscript to check for spelling errors.

All changes are marked in red in the revised manuscript.

We hope the revised manuscript will be suitable for publication in BMC Sports Science, Medicine and Rehabilitation.
Your sincerely,

Beat Knechtle

Enclosure: Manuscript via central website

Editorial comments

Judging by the reviewer’s comments, we anticipate a positive outcome for your manuscript; however, some minor language corrections are required before accepting it for publication. Please carefully copy-edit your study in order to avoid minor mistakes. We look forward to receiving your revised manuscript by 16 May 2014. If you imagine that it will take longer to prepare please give us some estimate of when we can expect it.

Answer: We agree with the editor and checked again for spelling errors. All changes are marked in red.

Reviewer’s report

Title: Swimming performances in long distance open-water events with and without wetsuit

Version: 3

Date: 7 May 2014

Reviewer: Daniel Marinho

Reviewer’s report:
I believe the authors corrected every concern raised by the reviewers.

Level of interest: An article whose findings are important to those with closely related research interests

Quality of written English: Needs some language corrections before being published

Answer: We agree with the expert reviewer and checked again for spelling errors. All changes are marked in red.

Statistical review: No, the manuscript does not need to be seen by a statistician.

Declaration of competing interests: I declare that I have no competing interests.